\documentclass[draft,12pt]{article}
%\bibliographystyle{naturemag}
\usepackage{linenofix}
%\usepackage[superscript]{cite}
\usepackage[dvips]{graphicx}
%
% Also uncomment the following command to allow illustrations to print
% when using Draft:
\setkeys{Gin}{draft=false}
%

\usepackage{amsmath,amsfonts,amsthm,amssymb}
\usepackage{epsfig,bm,epsf,float} %
\usepackage{color} %
\usepackage{lscape}
\usepackage{natbib}
%\usepackage{subfigure} %to include subfigures as (a),(b),...
%\usepackage{soul} % "soul" is a package which provides hyphenatable
%\usepackage{times} %change font to times %\usepackage{multirow} %create tables of multiple rows/columns.

\textwidth = 6.5 in \textheight = 9 in \oddsidemargin = 0.1 in
\evensidemargin = 0.1 in \topmargin = 0.1 in \headheight = 0.1 in
\headsep = 0.1 in
\parskip = 0.2in
\parindent = 0.2in


%letterspacing
\renewcommand{\baselinestretch}{1.5}
\renewcommand{\thepage}{S\arabic{page}} 
\renewcommand{\thesection}{S\arabic{section}}  
\renewcommand{\thetable}{S\arabic{table}}  
\renewcommand{\thefigure}{S\arabic{figure}} 
\renewcommand{\theequation}{S\arabic{equation}} 
\renewcommand{\figurename}{Supplemental Material, Figure} 
%% ------------------------------------------------------------------------ %%
%
% ENTER PREAMBLE
%
%% ------------------------------------------------------------------------ %%
%\textbf{Supplementary Information}

\title{Numerical simulations on megathrust rupture stabilized under strong dilatancy strengthening in slow slip region}
%\date{November 7, 2012}
\author{Yajing Liu\\ Department of Earth and Planetary Sciences, McGill University \\
Department of Geology and Geophysics, Woods Hole Oceanographic Inst.}

%90 character

%% Notice placement of commas and superscripts and use of &
%% in the author list

%\author{Yajing Liu$^{1}$}


\begin{document}
\begin{thebibliography}{9}
\providecommand{\natexlab}[1]{#1}
\expandafter\ifx\csname urlstyle\endcsname\relax
  \providecommand{\doi}[1]{doi:\discretionary{}{}{}#1}\else
  \providecommand{\doi}{doi:\discretionary{}{}{}\begingroup
  \urlstyle{rm}\Url}\fi

\bibitem[{\textit{Audet et~al.}(2009)\textit{Audet, Bostock, Christensen, and
  Peacock}}]{audet2009}
Audet, P., M.~G. Bostock, N.~I. Christensen, and S.~M. Peacock (2009), {Seismic
  evidence for overpressured subducted oceanic crust and megathrust fault
  sealing}, \textit{Nature}, \textit{457}, 76--78, \doi{10.1038/nature07650}.

\bibitem[{\textit{He et~al.}(2007)\textit{He, Wang, and Yao}}]{he2007}
He, C., Z.~Wang, and W.~Yao (2007), Frictional sliding of gabbro gouge under
  hydrothermal conditions, \textit{Tectonophys.}, \textit{445}, 353--362,
  \doi{10.1016/j.tecto.2007.09.008}.

\bibitem[{\textit{Hirose and Obara}(2005)}]{hirose2005}
Hirose, H., and K.~Obara (2005), {Repeating short- and long-term slow slip
  events with deep tremor activity around the Bungo channel region, southwest
  Japan}, \textit{Earth. Planets. Space.}, \textit{57}(10), 961--972.

\bibitem[{\textit{Liu and Rice}(2007)}]{liu2007}
Liu, Y., and J.~R. Rice (2007), Spontaneous and triggered aseismic deformation
  transients in a subduction fault model, \textit{J. Geophys. Res.},
  \textit{112}(B09404), \doi{10.1029/2007JB004930}.

\bibitem[{\textit{Liu and Rubin}(2010)}]{liu2010}
Liu, Y., and A.~M. Rubin (2010), {Role of fault gouge dilatancy on aseismic
  deformation transients}, \textit{J. Geophys. Res.}, \textit{115}(B10414),
  \doi{10.1029/2010JB007522}.

\bibitem[{\textit{Miyazaki et~al.}(2006)\textit{Miyazaki, Segall, Mcguire,
  Kato, and Hatanaka}}]{miyazaki2006}
Miyazaki, S., P.~Segall, J.~J. Mcguire, T.~Kato, and Y.~Hatanaka (2006),
  {Spatial and temporal evolution of stress and slip rate during the 2000 Tokai
  slow earthquake}, \textit{J. Geophys. Res.}, \textit{111}(B3).

\bibitem[{\textit{Peacock et~al.}(2002)\textit{Peacock, Wang, and
  McMahon}}]{peacock2002}
Peacock, S.~M., K.~Wang, and A.~M. McMahon (2002), Thermal structure and
  metamorphism of subducting oceanic crust: Insight into cascadia intraslab
  earthquakes, in \textit{The Cascadia subduction zone and related subduction
  systems: Seismic structure, Intraslab earthquakes and processes, and
  earthquake harzards}, edited by S.~Kirby, K.~Wang, and S.~Dunlop, pp.
  123--126, U. S. Geol. Surv. Open-File Rep., 02-328 and Geol. Surv. Canada
  Open-File, 4350.

\bibitem[{\textit{Rice}(1993)}]{rice1993}
Rice, J.~R. (1993), Spatiotemporal complexity of slip on a fault, \textit{J.
  Geophys. Res.}, \textit{98}(B6), 9885--9907.

\bibitem[{\textit{Sekine et~al.}(2010)\textit{Sekine, Hirose, and
  Obara}}]{sekine2010}
Sekine, S., H.~Hirose, and K.~Obara (2010), {Along-strike variations in
  short-term slow slip events in the southwest Japan subduction zone},
  \textit{J. Geophys. Res.}, \textit{115}(B00A27), \doi{10.1029/2008JB006059}.

\end{thebibliography}


\maketitle
\begin{center}
{\large \textbf{Supplement Information}}

{\large \textbf{Manuscript number: 2013GL055606}}
\end{center}
\listoftables
\listoffigures
%\begin{affiliations}
 %\item Department of Earth and Planetary Sciences, McGill University, Montreal, Canada
% \item Department of Geology and Geophysics, Woods Hole Oceanographic Institution, Woods Hole, Massachusetts, USA
%\end{affiliations}

%% ------------------------------------------------------------------------ %%
%
% TITLE
%
%% ------------------------------------------------------------------------ %%

%\affil{}
%% ------------------------------------------------------------------------ %%
%
% ABSTRACT
%
%% ------------------------------------------------------------------------ %%
% >> Do NOT include any \begin...\end commands within
% >> the body of the abstract.

\section{Description of model parameters}

The depth distributions of friction parameters follow our previous studies of slow slip events in a Cascadia-like 2-D subduction fault model \citep{liu2010}. The temperature-dependent gabbro friction stability parameter $a-b$ \citep{he2007} show large scatter, but to the first degree can be approximated by piecewise linear segments and mapped to the thrust fault using a Cascadia geotherm \citep{peacock2002}. As shown in main text Fig. 1, the velocity-weakening (potentially seismogenic) section consists of $a-b=-0.0035$ at intermediate depths to downdip $\sim 95$ km followed by a gradual increase to neutral stability ($a-b=0$) at $\sim$ 180 km. The fault slips at velocity-strengthening condition further downdip. At shallow depth, the effective normal stress increases linearly at a lithostatic minus hydrostatic pressure gradient of 18 MPa/km, $\bar\sigma = 18z + 2.8$, where vertical depth $z$ is in kilometers and $\bar\sigma$ has the unit of MPa. A non-zero normal stress of 2.8 MPa is set at the trench to represent the pressure due to sediments. $\bar\sigma$ remains a constant of 50 MPa for $z \geq 2.62$ km ($18z + 2.8 \geq 50$), except in the slow slip event region, where $\bar\sigma$ is set to a much lower level as near-lithostatic pore pressure condition has been suggested by several lines of evidence in the ETS areas \citep{liu2007,audet2009}. The SSE $\bar\sigma$ is chosen to be in a range between 1 and 5 MPa; individual values are presented for the simulation cases in this study. In the high effective normal stress seismogenic zone, the characteristic slip distance $d_{c}$ is defined by setting a constant large ratio between the nucleation zone size $h^* = 2 \mu d_c /[\pi \nu (b-a)_{max} \bar\sigma]$ \citep{rice1993} and the discretized grid size $h$, where $(b-a)_{max}=0.0035$. $h^*/h=16$ is used in all presented cases. With parameters as shown in Table \ref{tab:parameter}, $d_c = 11$ mm in the $\bar\sigma=50$ MPa zone. In the slow slip region, $d_c$ is correspondingly smaller; a typical value is 0.2 mm for all cases with dilatancy and 0.1 mm for the reference model without dilatancy. 

\begin{table}
\begin{center}
\begin{tabular}{lll}
\hline
Notation & Definition & Value [Unit]  \\ 
\hline
$(b-a)_{max}$ & maximum velocity-weakening value & 0.0035 \\
 $f_0$ & Nominal friction & 0.6 \\
 $W$ & Total fault downdip distance & 364 [km] \\
$h$ & Model grid size &  0.1 [km] \\ 
$h^*$ & Characteristic nucleation size & 16h = 1.6 [km]   \\
$\mu$ & Shear modulus & 30 [GPa]  \\
$\nu$ & Poisson's ratio & 0.25  \\
$V_0$ & Reference slip velocity &  $10^{-6}$ [m/s] \\ 
$V_{pl}$ & Plate convergence rate & 37 [mm/yr]\\
$\delta$ & Fault dipping angle & 12$^o$ \\

\hline

\end{tabular}
\vspace{0.1in}
\end{center} \label{tab:parameter}

\caption{Typical model parameters that are constant in all simulations cases. }
\end{table}

\section{Intermediate to undrained conditions}

As discussed by \citet{liu2010}, the pore pressure evolution with slip can be characterized by two dimensionless parameters: (1) a drainage parameter $U \equiv t_p/(d_c/V_{pl})$, the relative time scales for pore fluid diffusion and friction evolution, and (2) a dilatancy parameter $E \equiv  f_0(\epsilon/\beta)/(b \bar\sigma_0)$, the relative strength changes due to pore dilation and friction evolution, where $\bar\sigma_0 = \bar\sigma - p_0$ is the background effective normal stress.  When $U \gg 1$, the gouge is nearly undrained and there is no change in fluid mass. When $U \ll 1$, the gouge is nearly drained near slip rate $V_{pl}$, and there is no change in pore pressure. On a completely drained fault, $p$ instantaneously equilibrates with the ambient $p_0$ and solutions are essentially identical to the system without dilatancy (as in the Fig. 2 reference model). Thus, this study concerns intermediate drainage to nearly undrained conditions, which is investigated by using $U$ in the range of 0.01 to 100. That is, at $V_{pl}=37$ mm/yr, the characteristic pore pressure across-fault diffusion time $t_p$ is between $3\times10^{-3}$ to 30 years in the updip seismogenic zone where $d_c \approx 11$ mm. And $t_p$ is between $5.5\times 10^{-5}$ and $0.55$ years in the slow slip region where $d_c = 0.2$ mm. Note that under the quasi-dynamic approximation, the duration of modeled megathrust ruptures is typically between 400 and 600 seconds ($\sim 2 \times 10^{-5}$ years). 

As shown in Fig. \ref{fig:e1_Usz}, when the drainage parameter $U$ is varied between 0.01 and 100 independently in the updip seismogenic zone and the slow slip region, the median values of the maximum coseismic slip and downdip rupture limit are nearly independent of the choices of $U$. For each simulation case, the  modeled earthquake sequences have a median maximum coseismic slip very close to its no-dilatancy level. And the rupture stops slightly updip from the no-dilatancy termination depth. The range of variation appears to be smaller as the seismogenic zone becomes more undrained (increasing $U$). 

%\begi=n{landscape}
\begin{figure}
\begin{center}
\includegraphics[width=6.5 in]{figure/suppl/figS1.pdf}
\end{center}
\caption{Dependence of maximum coseismic slip and downdip rupture limit (top and bottom panel respectively in each subfigure) on the drainage parameter in the updip seismogenic zone ($U_{SZ}$) and SSE region ($U_{SSE}$). (a) $U_{SZ}=0.01$, (b) 0.1, (c) 1.0, (d) 10.0. $U_{SSE}$ varies from 0.01 to 10 for each fixed $U_{SZ}$. For each simulation case, thin black bar shows the variation between 10 to 90 percentile, thick black bar shows 25 to 75 percentile, red circle shows the median value. Horizontal dashed lines show the levels of the two source properties without dilatancy.} \label{fig:e1_Usz}
\end{figure}
%\end{landscape}

\section{Dilatancy effect on earthquake cycles} %to modify!
Incorporation of dilatancy affects a range of deformation modes in an earthquake cycle. Stress on the fault is perturbed which leads to large variations in the coseismic slip and interseismic period between individual earthquakes. As shown in Fig. \ref{fig:cycle}, following an earthquake (EQ1) of a similar size to that in the reference model, earthquakes EQs 2-4 have smaller coseismic slips and slower rupture speeds. In particular, EQ 4 is the only event in the sequence that does not break to the trench. In order to release the strain that is continuously accumulating but not completely released in previous cycles, the fault eventually breaks in a giant earthquake (EQ 5) with faster rupture speeds and a maximum coseismic slip of nearly 25 meters. Besides small SSEs in the low $\bar\sigma$ region, a large aseismic slip event emerges. These are unsuccessful attempts of earthquake nucleation that have slip velocities much higher than the small SSEs but not dynamic enough to reach the seismic threshold. In the cases shown in Fig. \ref{fig:cycle}, each interseismic period contains such a large aseismic event, which ``ruptures" not only the high pore pressure region but propagates further into the updip seismogenic zone and results in 1 to 2 meters of slip. 

\begin{figure}\begin{center}
\includegraphics[width=3.25 in]{figure/suppl/figS2.pdf}
\end{center}
\caption{(a) Slip history and (b) slip budget, with low dilatancy $\epsilon/\beta=0.1$ MPa at all depths. The five earthquakes, EQs 1-5, have maximum coseismic slips of 14, 10.8, 10.3, 4.8 and 23.9 meters and intervals from the respective preceding earthquake are 323, 281, 279, 262 and 454 years. The  slip budget is calculated for a cycle that consists of interseismic slip prior to EQ1, coseismic and postseismic slip of EQ1 in (a).} \label{fig:cycle}
\end{figure}


\section{Extremely small and large dilatancy effect cases on Figure 4 phase diagram}

Figs. \ref{fig:e0025e005} and \ref{fig:e0175e05} show the evolution of slip and slip rate for the modeled 3000-year history, for two cases representing extremely small ($\epsilon/\beta$ of 0.025 MPa in the updip seismogenic zone and 0.05 MPa in the downdip slow slip region) and large ($\epsilon/\beta$ of 0.175 MPa and 0.5 MPa, respectively, in the two regions) dilatancy effects. The two cases are also shown by white crosses in Fig. 4c phase diagram in the main text. 

In Fig. \ref{fig:e0025e005}, except for the first two megathrust events, small and large earthquakes occurred in pairs. The small earthquakes ruptured only a small portion of the fault; approximately between downdip 80 and 200 km for EQ 1, and between 100 and 185 km for EQs 3 and 5. The maximum coseismic slip is small as well, reaching 3.4, 2.2 and 1.5 meters, respectively, during EQs 1, 3 and 5. Results from these partial fault rupturing events are not included in the variational range and median value calculations in main text Fig. 4. The large earthquakes of the pairs, on the other hand, have maximum coseismic slips much greater than the no-dilatancy level, reaching 23.9, 22.0 and 18.7 meters, respectively, during EQs 2, 4 and 6. Updip rupture propagates all the way to the trench, and the downdip rupture limit is also slightly deeper than the no-dilatancy value. Such exceptionally large coseismic slip and spatial rupture extent thus contribute to Regime I in Figs. 4a and 4c phase diagrams where the two source characteristics go above the no-dilatancy level. 

In Fig. \ref{fig:e0175e05}, due to the strong dilatancy-strengthening effect, the coseismic rupture process usually breaks into several sub-events, each separated by several months of average slip rates 100 to 1000 times faster than the plate loading rate. Take the final rupture sequence in Fig. \ref{fig:e0175e05} for example, the first sub-event EQ 1 ruptured downdip approximately 60 to 110 km, accumulating $\sim 1.9$ meters of coseismic slip; EQ 2 nucleated where EQ 1 updip rupture stopped and propagated bilaterally updip to the trench and downdip to the SSE region with a maximum slip of 8.5 meters, the largest among this sequence; nucleation of EQ 3 is slightly further updip, due to the stress transfer, and rupture took place at the same spatial extent as EQ 2 but with a smaller amount of slip 6.2 meters. The intervals between the subsequent events are 104 days and 186 days. Modest slip accumulated during the inter-event periods due to the higher than $V_{pl}$ slip rate; for example, more than 4 meters of slip is released aseismically at the trench between EQ 2 and EQ 3. 

For the calculation of the percentile range and median value of the two earthquake characteristics in the main text and Supplement, only megathrust events with a maximum coseismic slip greater than 4 meters are included. The 4 m threshold was optimized choice such that (1) it is large enough to exclude earthquakes which only rupture a small portion of the megathrust fault, and (2) it is small enough to include small earthquakes produced at high $(\epsilon/\beta)_{SZ}$, such as 0.175 MPa as shown in Fig. S3. Those events do rupture the entire VW zone, but with a lot of preseismic and postseismic slip accumulation. 

\begin{figure}\begin{center}
\includegraphics[width=6 in]{figure/suppl/figS3.pdf}
\end{center}
\caption{Extremely small dilatancy effect case: $(\epsilon/\beta)_{SZ}=0.025$ MPa in the updip seismogenic zone, $(\epsilon/\beta)_{SSE}=0.05$ MPa in the slow slip event region between downdip 140 and 215 km. Drainage parameter is $U=1.0$ uniformly along the entire fault. (a) Slip history on the fault. Black lines are interseismic slip every 50 years. Red lines are coseismic slip every 20 seconds.  A median coseismic slip of 22 meters is calculated for earthquakes with slip larger than 4 m (here, 24.8, 15.5, 23.9, 22.0, 18.7 meters respectively), and the median of their rupture downdip limits is 238.65 km. This is further downdip than the no-dilatancy case, due to larger coseismic slip in general. (b) Maximum slip rate on the fault in $\log_{10}(V_{max}/V_{pl})$. Vertical dashed line represents the cutoff threshold for ``coseismic" slip. (c) Maximum slip rate for a 50-yr SSE periods.}\label{fig:e0025e005}
\end{figure}

\begin{figure}\begin{center}
\includegraphics[width=6 in]{figure/suppl/figS4.pdf}
\end{center}
\caption{Extremely high dilatancy effect case: $(\epsilon/\beta)_{SZ}=0.175$ MPa in the updip seismogenic zone, $(\epsilon/\beta)_{SSE}=0.5$ MPa in the SSE region. $U=1.0$. (a) Slip history on the fault. Black lines are interseismic slip every 50 years. Red lines are coseismic slip every 20 seconds. A median coseismic slip of 7.7 meters is calculated for modeled earthquakes with slip larger than 4 meters (here, 7.5, 7.5, 9.1, 6.2, 8.6, 8.0, 8.5, 6.2 meters respectively). The median of their rupture downdip limits is 142.4 km. (b) Maximum slip rate on the fault in $\log_{10}(V_{max}/V_{pl})$. Vertical dashed line represents the cutoff threshold for ``coseismic" slip. (c) Maximum slip rate during a 3-yr period containing the final rupture sequence EQ 1 to EQ 3. 
}\label{fig:e0175e05}
\end{figure}
\clearpage
\section{Long-term and short-term SSEs}
As discussed in the main text, the transition from shot-term to long-term SSEs behavior as dilatancy effect increases (Fig. 4) provides insights to constructing a model that can produce both types of SSEs simultaneously with properties similar to those observed in Tokai and Bungo Channel, SW Japan \citep{hirose2005,miyazaki2006,sekine2010}. Fig. \ref{fig:ltst} shows such an example. The SSE zone is defined from downdip 110 to 215 km, under a low $\bar\sigma=1$ MPa and $d_c=0.05$ mm. Its updip portion (110-145 km) is under strong dilatancy $\epsilon/\beta=0.5$ MPa, and downdip portion (145-215 km) is under low dilatancy $\epsilon/\beta=0.01$ MPa, and the rest of the fault has $\epsilon/\beta=0.1$ MPa. Drainage parameter $U=1$ at all depths. As shown in Fig. \ref{fig:ltst}a, short-term SSEs repeat roughly every 0.3 year, with an average duration of 1-2 weeks, and slip is mostly limited to 145-200 km. Long-term SSEs repeat roughly every 2.5 years, with an average duration of 0.3 years, and slip takes place 110-200 km. More interestingly, short-term and long-term SSEs may occur simultaneously on the fault. The relative spatiotemporal distribution of the two types of SSEs are qualitatively similar to those observed in the Bungo Channel, Nankai Trough, although extensive parameter space search is needed in order to quantitatively reproduce the frequency and slip amplitude observations \citep{hirose2005}.

\begin{figure}
\begin{center}
\includegraphics[width=4 in]{figure/suppl/figS5.pdf}
\end{center}
\caption{An example of spatiotemporal relative distribution of short-term and long-term SSEs under heterogenous dilatancy conditions.}\label{fig:ltst}
\end{figure}


\clearpage 
%\bibliography{bibliu} 
%\bibliographystyle{agu04}



%%% End of body of article:
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% Optional Appendix goes here
%
%%%%%%%%%%%%%%%%%
% Geophysical Research Letters only allows an appendix without a letter.
%% You can get this result with
% \section*{Appendix}
% or
% \section*{Appendix: Title}
%%%%%%%%%%%%%%%%%
%
% \appendix resets counters and redefines section heads
% but doesn't print anything.
% After typing \appendix
%
% \section{Here Is Appendix Title}
% will print
% Appendix A: Here Is Appendix Title
%
% \section*{Appendix}
% will print
% Appendix
%
% \section*{Appendix: Here Is Appendix Title}
% will print
% Appendix: Here Is Appendix Title
%
% For only 1 appendix \appendix \section{Appendix} is preferred.
% which will print
% Appendix A
%\end{end}
%\appendix
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Optional Glossary or Notation section, goes here
%
%%%%%%%%%%%%%%
% Glossary only allowed in Reviews of Geophysics
% \section*{Glossary}
% \paragraph{Term}
% Term Definition here
%
%%%%%%%%%%%%%%
% Notation -- End each entry with a period.
% \begin{notation}
% Term & definition.\\
% Second Term & second definition.
% \end{notation}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% ACKNOWLEDGMENTS
%\begin{acknowledgments}

%\end{acknowledgments}
%% ------------------------------------------------------------------------ %%
%
% REFERENCE LIST AND TEXT CITATIONS
%
% Either type in your references using
% \begin{thebibliography}{}
% \bibitem
% Text
% \end{thebibliography}
%
% Or,
%
% If you use BiBTeX for your References, please produce your .bbl
% file and copy the contents into your paper here.
%
% Follow these steps:
% 1. Run LaTeX on your LaTeX file.
%
% 2. Run BiBTeX on your LaTeX file.
%
% 3. Open the new .bbl file containing the reference list and
% copy all the contents into your LaTeX file here.
%
% 4. Comment out the old \bibliographystyle and \bibliography commands.
%
% 5. Run LaTeX on your new file before submitting.
%
% AGU does not want a .bib or a .bbl file, but asks that you
% copy in the contents of your .bbl file here.
%\begin{thebibliography}{}
%\bibitem{jskilby}J. S. Kilby,
%``Invention of the Integrated Circuit,'' {\it IEEE Trans. Electron Devices,}
%{\bf ED-23,} 648 (2008).
%\end{thebibliography}
%\bibliography{bibliu}
%\bibliographystyle{nature}
%% ------------------------------------------------------------------------ %%
%
% END ARTICLE
%
%% ------------------------------------------------------------------------ %%
%\end{article}
\clearpage
%% Enter Figures and Tables at here:

% When submitting articles through the GEMS system:
% COMMENT OUT ANY COMMANDS THAT INCLUDE GRAPHICS.
% Figure captions go below this illustration; Table captions above tables
% ONE-COLUMN figure/table, including eps graphics
%  

% \end{document}
%
% \begin{table}
% \caption{}
% \end{table}
%
% ---------------
% TWO-COLUMN figure/table
%
% \begin{figure*}
% \noindent\includegraphics[width=39pc]{samplefigure.eps}
% \caption{Caption text here}
% \end{figure*}
%
% \begin{table*}
% \caption{Caption text here}
% \end{table*}
%
% see below for how to make landscape figures or tables
\end{document}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
More information and Advice:
% SECTION HEADS
---------------
Level 1 head
Use the \section{} command to identify level 1 heads;
type the appropriate head wording between the curly
brackets, as shown below.
Capitalize the first letter of each word (expect for
prepositions, conjunctions, and articles that are
three or fewer letters).
Do not hyphenate level 1 heads. To break lines,
type \protect\\ where you want the break to occur.
AGU prefers the inverted triangle, breaking before
prepositions, conjunctions, and articles, if possible.
An example:
\section{Level 1 Head: Introduction}
---------------
Level 2 head
Use the \subsection{} command to identify level 2 heads.
Capitalize the first letter of each word (expect for
prepositions, conjunctions, and articles that are
three or fewer letters).
Do not hyphenate level 1 heads. To break lines,
type \protect\\ where you want the break to occur.
AGU prefers the inverted triangle, breaking before
prepositions, conjunctions, and articles, if possible.
\subsection{Level 2 Head} An example.
---------------
Level 3 head
Use the \subsubsection{} command to identify level 3 heads
Capitalize only the first letter of the first word, acronyms,
first letter of proper nouns, and first letter of first word
after a colon.
Hyphenation is permitted in level 3 heads, if needed.
\subsubsection{Level 3 Head} An example.
\subsubsubsection{Level 4 Head} An example.
%% ------------------------------------------------------------------------ %%
%
% IN-TEXT LISTS
%
%% ------------------------------------------------------------------------ %%
% Do not use bulleted lists; enumerated lists are okay.
begin{enumerate}
\item
\item
\item
\end{enumerate}
%% ------------------------------------------------------------------------ %%
%
% EQUATIONS
%
%% ------------------------------------------------------------------------ %%
% Single-line equations are centered.
Math coded inside display math mode \[ ...\]
will not be numbered e.g.:
\[ x^2=y^2 + z^2\]
Math coded inside \begin{equation} and \end{equation} will
be automatically numbered e.g.:
\begin{equation}
x^2=y^2 + z^2
\end{equation}
% IF YOU HAVE MULTI-LINE EQUATIONS, PLEASE
% BREAK THE EQUATIONS INTO TWO OR MORE LINES
% OF SINGLE COLUMN WIDTH (20 pc, 8.3 cm)
% using double backslashes (\\).
% To create multiline equations, use the
% \begin{eqnarray} and \end{eqnarray} environment
% as demonstrated below.
\begin{eqnarray}
x_{1} & = & (x - x_{0}) \cos \Theta \nonumber \\
&& + (y - y_{0}) \sin \Theta \nonumber \\
y_{1} & = & -(x - x_{0}) \sin \Theta \nonumber \\
&& + (y - y_{0}) \cos \Theta.
\end{eqnarray}
If you don't want an equation number, use the star form:
\begin{eqnarray*}...\end{eqnarray*}
% Break each line at a sign of operation
% (+, -, etc.) if possible, with the sign of operation
% on the new line.
% Indent second and subsequent lines to align with
% the first character following the equal sign on the
% first line.
% Use an \hspace{} command to insert horizontal space
% into your equation if necessary. Place an appropriate
% unit of measure between the curly braces, e.g.
% \hspace{1in}; you may have to experiment to achieve
% the correct amount of space.
% There is another multiline equation environment:
% \begin{aguleftmath}...\end{aguleftmath}
% The equation is aligned left and the second line indents to
% the width of a paragraph indent (AGU style)
%% ------------------------------------------------------------------------ %%
%
% EQUATION NUMBERING: COUNTER
%
%% ------------------------------------------------------------------------ %%
% You may change equation numbering by resetting
% the equation counter or by explicitly numbering
% an equation.
% To explicitly number an equation, type \eqnum{}
% (with the desired number between the brackets)
% after the \begin{equation} or \begin{eqnarray}
% command. The \eqnum{} command will affect only
% the equation it appears with; LaTeX will number
% any equations appearing later in the manuscript
% according to the equation counter.
%
% To reset the equation counter, place the setcounter{equation}
% command in front of your equation(s).
%\setcounter{equation}{0}
% Set the equation counter to 0 if the next
% number needed is 1 or set it to 7 if the
% next number needed is 8, etc.
%
% The \setcounter{equation} command does affect
% equations appearing later in the manuscript.
% If you have a multiline equation that needs only
% one equation number, use a \nonumber command in
% front of the double backslashes (\\) as shown in
% the multiline equation above.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% Landscape figure and table examples
%
% ---------------
% Landscape (broadside) figure/table
% (These objects will not display properly in draft mode, use galley.)
%
% ONE-COLUMN landscape figure and table
%
% \begin{landscapefigure}
% \includegraphics[height=.75\mycolumnwidth,width=42pc]{samplefigure.eps}
% \caption{Caption text here}
% \end{landscapefigure}
%
% \begin{landscapetable}
% \caption{Caption text here}
% \begin{tabular*}{\hsize}{@{\extracolsep{\fill}}lcccc}
% \tableline
% ....
% \tableline\\
% \multicolumn5l{(a) Algorithms from Numerical Recipes}\\
% \end{tabular*}
% \tablenotetext{}{}
% \tablecomments{}
% \end{landscapetable}
%
% FULL-PAGE landscape figures and tables
%
% \begin{figure*}[p]
% \begin{landscapefigure*}
% illustration here
% \caption{caption here}
% \end{landscapefigure*}
% \end{figure*}
%
% \begin{table}[p]
% \begin{landscapetable*}
% \caption{}
% \begin{tabular*}{\textheight}{@{\extracolsep{\fill}}lccrrrcrrr}
% ....
% \end{tabular*}
% \begin{tablenotes}
% ...
% \end{tablenotes}
% \end{landscapetable*}
% \end{table}
%

